Author’s response to reviews

Title: Motivational Interviewing as an Intervention to Increase Adolescent Self-Efficacy and Promote Weight loss

Authors:

Beverly Walpole (bwalpole@uoguelph.ca)
Elizabeth Dettmer (elizabeth.dettmer@sickkids.ca)
Barbara Morrongiello (bmorrong@uoguelph.ca)
Brian McCrindle (brian.mccrindle@sickkids.ca)
Jill Hamilton (jill.hamilton@sickkids.ca)

Version: 2 Date: 13 April 2011

Author’s response to reviews:

Revisions made to manuscript-
1. In the "Methods" section, the addition of ethical approval (institution names and reference numbers)
2. In the "Acknowledgments" section, the addition of acknowledgment of the primary author's personal doctoral award as support of funding
* note- an email has been sent to the journal's editorial office with proof of external funding.